Author's response to reviews

Title: Design of the Advance Directives Cohort: a study of end-of-life decision-making focusing on Advance Directives

Authors:

Matthijs PS van Wijmen (m.vanwijmen@vumc.nl)  
Mette L Rurup (m.rurup@vumc.nl)  
H Roeline W Pasman (hrw.pasman@vumc.nl)  
Pam J Kaspers (p.kaspers@vumc.nl)  
Bregje D Onwuteaka-Philipsen (b.philipsen@vumc.nl)

Version: 9 Date: 23 March 2010

Author's response to reviews:

Dear Editor,

We re-submit the main document of our manuscript.

The changes we made concern both tables in the main document:

- In table 1 we shifted the research questions in the second column, so that they are all at the same position on the page underneath each other.

- When it comes to table 2, you made some changes concerning the title and the legend of the table. It seemed you did not want that title and legend were part of the table itself. As a result of this, however, part of the legend was above and part was underneath the table. We made some changes so that title and legend are not part of the table, but all text is in the right place.

Yours sincerely,

Also on behalf of the co-authors

Matthijs van Wijmen